\begin{figure*}\label{fig:complexity}
\begin{center}
\begin{tabular}{l||l|l|l|l|l|l|l|}
\cline{2-8}
\textbf{Runtime while}&Naive Bayes&Ridor             &Clump               &LWL      &j48/C4.5          &RIPPER              &OneR   \\ \hline \hline
Training              & O$(nk)$   & O$(nk * log(kn)$ & O$(3nk * log(kn))$ & O$(n)$  & O$(nk * log(n))$ & O$(2nk * log(kn))$ &O$(kn)$\\ \hline
Testing               & O$(mk)$   & O$(md)$          & O$(mk)$ - O$(mkd)$ & O$(mnk)$& O$(m * log(m))$  & \textit{Unlisted}  &O$(1)$ \\ \hline
\end{tabular}

\caption{Runtime Complexity of Clump, Naive Bayes\cite{NBComplexity}, Ridor, OneR\cite{OneRComplexity}, LWL\cite{LWL}, j48/C4.5\cite{C4.5Complexity}, and RIPPER\cite{RIPPERComplexity} on a dataset with \textit{n} training cases, \textit{m} testing cases, \textit{d} tree depth, and \textit{k} attributes.}
\end{center}
\end{figure*}
